A comparison of regional climate projections with a range of climate sensitivities

Clair Barnes¹, Richard E Chandler², and Christopher Metcalfe Brierley³

¹Imperial College London ²UCL ³University College London

December 19, 2023

A comparison of regional climate projections with a range of climate sensitivities

Clair R. Barnes^{1,2}, Richard E. Chandler¹ and Christopher M. Brierley³

¹Department of Statistical Science, University College London, Gower Street, London WC1E 6BT ²Grantham Institute, Imperial College London, Exhibition Road, London SW7 2AZ ³Department of Geography, University College London, Gower Street, London WC1E 6BT

Key Points:

1

2

3

4 5 6

7

13

8	•	The UKCP and EuroCORDEX regional model ensembles have similar biases, but
9		project very different future climate over the UK
10	•	These differences are driven largely by differences in the climate sensitivity of the
11		GCMs used to force the regional models
12	•	Comparing projections after a specified degree of warming, rather than in spec-
13		ified decades, reduces but does not resolve these differences

Corresponding author: Clair Barnes, c.barnes22@imperial.ac.uk

14 Abstract

To investigate the extent to which differences in regional model projections can be ex-15 plained by differences in the warming rates of their driving models, we compare projec-16 tions of temperature and precipitation over the UK from two regional climate ensem-17 bles – the EuroCORDEX multi-model ensemble and UKCP18 perturbed parameter en-18 semble – along with projections produced by the 'parent' GCMs from which boundary 19 conditions were taken. We evaluate the ensembles in terms of their representation of re-20 cent climate, then compare the changes simulated between 1981–2010 and 2050–2079. 21 While both ensembles exhibit seasonal biases with similar magnitudes and spatial pat-22 terns during the evaluation period, the UKCP18 ensemble exhibits a somewhat stronger 23 change signal in future simulations, due to a combination of higher climate sensitivity 24 of the driving models, variations in the forcings applied, and - in the regional simulations 25 - the inclusion of time-varying aerosols. 26

In order to reconcile the two sets of projections, we compare two periods correspond-27 ing to fixed global warming levels in the driving models, to constrain the variability within 28 and between the ensembles which can be ascribed to differing rates of global warming: 29 the discrepancy between the ensembles is greatly reduced, although some differences in 30 the local response remain, with the UKCP18 runs slightly warmer and drier than the 31 EuroCORDEX runs, particularly in summer. We also highlight potential pitfalls of com-32 paring warming levels with a reference time period, due to uncertainty about the warm-33 ing that has already occurred in the driving models prior to the reference period. 34

³⁵ Plain Language Summary

We compare temperature and precipitation over the UK from two different collec-36 tions (known as 'ensembles') of climate model runs: the EuroCORDEX ensemble, con-37 sisting of simulations from many combinations of global- and regional-scale models; and 38 the UKCP18 regional ensemble, which uses a single pair of models, but adjusts the model 39 parameters for each run. Both ensembles perform well in the current climate, but future 40 changes in the UKCP18 ensemble are generally larger by 2050-2079 than those in the 41 EuroCORDEX ensemble. This is largely because the UKCP18 global models warm more 42 quickly in response to the greenhouse gases in the atmosphere, and use slightly higher 43 concentrations of greenhouse gases. 44

To understand the differences between the two ensembles that cannot be explained by differences in the rate of global warming, we look at changes as the models warm from 1°C to 2°C globally above levels in the early 20th century. This reduces the discrepancy between the ensembles, although some differences remain: the UKCP18 ensemble remains slightly warmer and drier than EuroCORDEX, particularly in summer. We highlight issues that arise when comparing simulations at a given warming level against simulations in a fixed decade, due to uncertainty about how much warming has already occurred.

52 1 Introduction

Adapting to climate change will be one of the great challenges of the twenty-first 53 century. Knowledge of how future changes will impact a locality is an important prereq-54 uisite to planning for them. Many available global climate projections do not provide 55 information at local spatial scales, and the use of regional climate models (RCMs) to dy-56 namically downscale their coarse resolution is an important step to provide locally-relevant 57 information (Jacob et al., 2014; Sørland et al., 2018). Uncertainty about the potential 58 range of future changes is often assessed through the use of ensembles (coherent collec-59 tions) of simulations (von Trentini et al., 2019; Lehner et al., 2020). Large model ensem-60 bles, consisting of simulations using the same climate models but initialised with differ-61 ent atmospheric conditions, aim to sample internal variability within a particular model 62

(Collins et al., 2006, 2011); whilst perturbed parameter ensembles (PPEs), in which mul-63 tiple simulations are again produced by a single model but now varying the parameters 64 controlling the representation of physical processes in each realisation, examine the ef-65 fects of the associated uncertainties. In contrast to this, a multi-model ensemble (MME) 66 in which all simulations use the same pathway of future emissions or atmospheric con-67 centrations, but different combinations of global and regional models, can also sample 68 the uncertainty arising from the formulation of the selected models (von Trentini et al., 69 2019; Christensen & Kjellström, 2020). Ensembles such as the global Coupled Model In-70 tercomparison Project Phase 5 experiment (Taylor et al., 2012, CMIP5) and, at a regional 71 scale, the CoOrdinated Regional Downscaling EXperiment (Giorgi & Gutowski Jr, 2015, 72 CORDEX) are essential tools to understand the potential range of future climate im-73 pacts. 74

Here we compare projections of temperature and precipitation from two regional 75 climate ensembles at the same spatial scale – the CORDEX MME (Jacob et al., 2014) 76 and the UK Climate Projections 2018 PPE (Murphy et al., 2018) – along with projec-77 tions produced by the global General Circulation Models (GCMs) used to drive the re-78 gional models. We note that the GCMs are not designed or expected to capture detailed 79 orographic or coastal effects, and so can only be fairly assessed at the national scale; how-80 ever, they are included here so that the contributions and capabilities of the downscal-81 ing RCMs can be evaluated. Both the EuroCORDEX and UKCP18 ensembles aim to 82 provide plausible climate projections under the RCP8.5 emissions scenario, but they project 83 quite different of outcomes by the late 21st century. To explore the differences between 84 the two ensembles, we compare projections not only between a reference period and 2050-85 2079, but also between global warming levels of 1° C and 2° C compared with the early 86 20th century. We focus on the climate of the United Kingdom, a relatively small region 87 for which decision-making and planning are frequently quite localised in comparison to 88 other, larger countries, and for which the spatial resolution of the GCMs is too coarse 89 to provide the local-scale information required for localised adaptation. However, out-90 puts from both of the regional ensembles are also available for a much wider area encom-91 passing most of Europe, and we anticipate that the considerations around the use of Global 92 Warming Levels (GWLs) with regional climate projections, along with the broader points 93 raised in the Discussion, will have broader relevance to users of those and other regional 94 ensembles. 95

The ensembles of simulations used in the study are described, along with the nec-96 essary preprocessing, in Section 2; the methods used to regrid the data and calculate cli-97 matologies are described in Sections 2.2 and 2.3. Section 2.4 describes the approach used 98 to calculate the GWL climatologies and highlights some important caveats to be con-99 sidered when using GWLs with regional climate model output. In Section 3 we evalu-100 ate the representation of historical and future surface temperatures in the ensembles; a 101 similar analysis is carried out for projected changes in seasonal precipitation in Section 102 4. Section 5 concludes with a discussion of potential benefits and drawbacks of the use 103 of GWLs. 104

Any plots referred to but not shown in the main text can be found in the supplementary material, or — along with plots of other climate indices — by using the EuroCORDEX-UK Plot Explorer tool at https://github-pages.ucl.ac.uk/EuroCORDEX-UK-plot-explorer/ (Barnes et al., 2023).

109 2 Methods

110 2.1 Datasets

The analysis is focused mainly on a comparison of the regional component of UKCP18, the latest suite of national climate projections for the UK (Murphy et al., 2018), with

projections produced by the EuroCORDEX project under the RCP8.5 scenario (Van Vu-113 uren et al., 2011; Jacob et al., 2014). UKCP18 provides a range of different products, 114 the regional component of which is a 12-member PPE that uses HadREM3-GA7-05 to 115 downscale output from the global HadGEM3-GC3.05 model at approximately 60km res-116 olution to a resolution of 0.11° over Europe, equivalent to about 12km resolution over 117 the UK (Murphy et al., 2018). Each numbered ensemble member uses the same pertur-118 bations at both 60km and 12km resolutions, with the first ensemble member having no 119 perturbations from the standard model. The ensemble members additionally sample a 120 range of future emissions scenarios consistent with the single RCP8.5 pathway used in 121 the CMIP5 experiments, rather than using the RCP8.5 pathway directly. CO₂ pathways 122 were chosen to represent the range of outcomes indicated by the UKCP18 probabilistic 123 projections (Murphy et al., 2018), with most of the pathways falling above the standard 124 RCP8.5 scenario; in addition, some of the perturbed parameters relate to scalings of an-125 thropogenic aerosol emissions (Sexton et al., 2021; Yamazaki et al., 2021). Henceforth, 126 the ensemble of regional runs will be referred to as UKCP regional, and the global PPE 127 as GC3.05-PPE, in line with UKCP documentation. 128

At the time of writing, the EuroCORDEX project has produced runs driven by RCP8.5 129 forcings from six of the coupled ocean-atmosphere models run as part of the Coupled Model 130 Intercomparison Project Phase 5 (CMIP5) experiment (Taylor et al., 2012), using thir-131 teen RCMs (Jacob et al., 2014). However, runs have only been produced for a subset of 132 the possible GCM-RCM pairs, and the EuroCORDEX ensemble used in the present anal-133 ysis consists of 64 climate simulations, shown in Figure 1. Two of the GCMs provided 134 three independent realisations to the project, but each marked GCM-RCM pair contributes 135 a single run to the 64-member ensemble. The EuroCORDEX models are also run at 0.11° 136 resolution over Europe, with the exact spatial extent varying according to the downscal-137 ing RCM. Henceforth, the ensemble of runs used to drive the EuroCORDEX simulations 138 will be referred to as CMIP5-EC. 139

For each of the ten RCMs listed in Figure 1 the EuroCORDEX ensemble also pro-140 vides a single evaluation run forced by ERA-Interim reanalysis (Dee et al., 2011) rather 141 than by GCM output: these runs allow the performance of the RCMs to be evaluated 142 in the absence of errors or biases inherited from the driving GCMs. The evaluation pe-143 riods for which these runs were produced differ between models, with only the period 144 from January 1st 1989 – December 31st 2008 covered by all of the runs. Biases in the 145 model output during this period are evaluated against interpolated daily estimates of 146 historical precipitation and daily maximum and minimum temperature – referred to hence-147 forth as the observations – from the HadUK-Grid dataset (Hollis et al., 2019). Where 148 observations of daily mean temperature are required, the mean of the daily maximum 149 and minimum is used (Perry et al., 2009). 150

151

2.2 Regridding onto a common grid

The various ensembles considered in this paper include models run at different spatial resolutions (e.g. the outputs from the RCMs as well as from the GCMs used to drive them) and with different native grids. To facilitate direct comparison across all of the ensembles, each model's outputs are interpolated from the native grid onto a common grid. In this paper, all data are presented on the same 12km grid used in the HadUK-Grid data set and UKCP regional over the UK land surface.

Indices are first computed on each model's native grid, then interpolated to the 12km grid using a conservative area-weighting scheme (Jones, 1999). When regridding the regional model outputs, only grid cells falling within the UK land surface are used: this is to avoid introducing bias by interpolating across the land-sea boundary. When regridding the lower-resolution CMIP5-EC and GC3.05-PPE output however, this approach is not used: removing cells flagged as belonging to the sea surface before regridding the



Figure 1. The 64 GCM-RCM pairs included in the EuroCORDEX ensemble.

data would result in an absence of data in large areas of the UK. Instead, the low-resolution 164 data are regridded directly onto the land surface 12km grid, and the effect of any result-165 ing blurring of land and sea surface variables is highlighted when discussing the results 166 below. This choice was made to keep the focus of this paper on the 12km resolution of 167 the regional climate models; if GCM performance was of direct interest then it would 168 be more informative to compare the GCMs to observations on a coarser grid, for exam-169 ple the 60km version of HadUK-Grid (Hollis et al., 2019). However, as noted above, re-170 sults for GCMs are presented here primarily to illustrate which aspects of the RCM per-171 formance are largely inherited from the driving models, and which arise from the down-172 scaling models themselves. 173

174 **2.3** Calculating climatologies

Model biases are calculated as the difference between the model climatology and 175 the equivalent HadUK-Grid observed climatology during the common evaluation period 176 from January 1st 1989 to December 31st 2008. Changes in temperature-based indices 177 are calculated as the difference between the aggregated value of the index during the fu-178 ture period (December 1st 2049 – November 30th 2079) and the reference period (De-179 cember 1st 1980 – November 30th 2010). For precipitation indices, biases are presented 180 as relative (percentage) differences with respect to observed precipitation, and changes 181 as relative differences with respect to the reference climatology. UKCP18 users should 182 note that this is not the same reference period as that used in the original UKCP18 anal-183 ysis, which considered twenty-year periods (Murphy et al., 2018): instead, the present 184 paper focuses on the thirty-year time-slices recommended by the World Meteorological 185 Organisation (WMO, 2017). 186

2.4 Changes between global warming levels

187

The GCMs used to drive the EuroCORDEX and UKCP regional ensembles have very different climate sensitivities (Flato et al., 2014; Yamazaki et al., 2021) and, as noted in Section 2.1, the GC3.05-PPE runs also use a variant of the standard RCP8.5 emissions scenario. This translates into rather different rates of warming — illustrated in Table 1, which shows the change in GMST in each GCM between 1900–1950 and 2050–2079
 — which might be expected to propagate into systematic differences between the regional outputs, obscuring potentially interesting differences in local responses.

Table 1. GMST change (°C) from 1900-1950 to 2050–2079 in the CMIP5-EC and GC3.05-PPE runs. CMIP5-EC values are taken from the IPCC Interactive Atlas (Iturbide et al., 2021); GC3.05-PPE values were calculated directly from the area-weighted model output.

CMIP5-EC	GO	GC3.05-PPE			
GCM	$^{\circ}\mathbf{C}$	Member	$^{\circ}\mathbf{C}$	Member	$^{\circ}\mathbf{C}$
CNRM-CM5	2.7	01	3.9	09	4.3
EC-EARTH	2.9	04	4.2	10	3.7
HadGEM2-ES	3.4	05	3.7	11	3.9
IPSL-CM5A-MR	3.6	06	3.8	12	3.2
MPI-ESM-LR	3.1	07	3.6	13	3.8
NorESM1-M	2.6	08	3.3	15	3.8

One approach to controlling the variability associated with both the rates of warm-195 ing exhibited by different models and the choice of emissions scenario is to compare changes 196 in model climatology not at particular time periods but at periods centred on the year 197 in which the change in global mean surface temperature since preindustrial levels exceeds 198 a particular threshold of interest, known as the global warming level (GWL) (James et 199 al., 2017; Hausfather et al., 2022). This approach was adopted in the IPCC's AR6 (Chen 200 et al., 2021), and ensemble means of the CORDEX projections at specified GWLs are 201 available through the IPCC's Interactive Atlas (https://interactive-atlas.ipcc.ch). 202 By fixing the GWL in this way, inter-model variation arising from the choice of forcing 203 scenario and from differences between the driving models' global responses to greenhouse 204 gases is reduced: the remaining differences between the runs may therefore be attributed 205 with greater confidence to differences in the local climate response and natural variabil-206 ity (James et al., 2017). 207

Particular care must be taken when using the GWL approach to evaluate changes 208 in regional model output, although the authors are not aware of any case in the liter-209 ature where this has previously been highlighted. This is because, while GWLs are typ-210 ically calculated with respect to a preindustrial reference, regional climate model out-211 put is typically only available from the late twentieth century onward. As a result, changes 212 are commonly reported with respect to a reference period beginning no earlier than 1980: 213 for example, the IPCC Interactive Atlas presents changes of climate indices computed 214 from CORDEX regional model output at GWLs of 1.5, 2, 3 and 4°C with respect to three 215 reference periods beginning later than 1980 (1981–2010, 1986–2005, and 1995–2014). How-216 ever, due partly to the differences in climate sensitivity that GWL selection is intended 217 to mitigate, the driving runs have already warmed by different amounts between the prein-218 dustrial and reference periods. Figure 2a, showing the change in GMST between 1900– 219 1950 (used in place of a preindustrial baseline due to unavailability of earlier GC3.05-220 PPE output) and the reference period of 1981–2010, illustrates this. The observed GMST 221 increase during this period was approximately 0.6°C (calculated from HadCRUT.5.0, Morice 222 et al. (2021)), and as Figure 2a shows, more than half of the CMIP5-EC driving runs 223 have already exceeded this threshold before the start of the regional model output. In 224 contrast, all but one of the GC3.05-PPE runs have warmed by less than 0.5° C prior to 225 the reference period. This systematic difference can largely be attributed to GC3.05-PPE's 226 strong cooling response to increased aerosol concentrations during the second half of the 227 twentieth century (Murphy et al., 2018; Tucker et al., 2021), which may mean a strong 228 warming response to greenhouse gas forcing emerges during model development (Nijsse 229



Figure 2. Boxplots of annual temperature changes in the CMIP5-EC and GC3.05-PPE runs: (a) global temperature change from the early 20th century (1900–1950) to 1981–2010; and changes in mean UK land near-surface temperature from (b) 1981–2010 and (c) the early 20th century to the 30-year time period centred on the year in which the driving model exceeded a 2°C increase in GMST with respect to early 20th century climate.

et al., 2020). While the difference is not particularly problematic in the EuroCORDEX 230 ensemble (in part due to the relatively low warming rates of most of the driving mod-231 els), preliminary analysis suggests that the CMIP6 models simulate an even wider spread 232 of historical changes, ranging from -0.05 to 1.06 degrees: it is therefore very that the range 233 of temperature changes observed in any representative CMIP6-driven CORDEX ensem-234 ble prior to the reference period would be somewhat wider than in the current EuroCORDEX 235 and UKCP regional ensembles, although we note that a balanced ensemble design is planned 236 for the next CORDEX phase to sample the range of climate sensitivities in CMIP6 more 237 systematically (Sobolowski et al., 2023). 238

The potential for confusion caused by comparing a GWL with a fixed reference pe-239 riod is illustrated in Figure 2b, which shows the change in UK mean temperature in each 240 of the driving models between the reference period (1981–2010) and the year in which 241 each model's GMST first exceeded 2°C. Consideration only of the changes between the 242 reference period and a particular GWL in this way fails to take into account the effect 243 of the models' differing warming rates prior to the reference period: the GMST of each 244 model during the reference period is unknown, and as a result, it is not clear how to in-245 terpret the changes. Furthermore, because the GC3.05-PPE runs were generally cooler 246 during the reference period than the CMIP5-EC ensemble (Figure 2a), UK temperatures 247 appear to have warmed somewhat more in GC3.05-PPE by the time the models reach 248 2° C than in CMIP5-EC. In Figure 2c, projected changes are instead evaluated against 249 a reference GWL – here, the early twentieth century. This removes almost all variabil-250 ity due to climate sensitivity and to the choice of forcing scenario, leaving only the mod-251 els' regional response to a defined period of global warming: once this source of variabil-252 ity is accounted for, the local responses of the two ensembles are in fact fairly similar. 253 This makes interpretation of Figure 2c straightforward: in both ensembles, most of the 254 runs simulate between 1.5 and 2°C of warming over the UK land surface in response to 255 a 2°C change in GMST, with ensemble mean changes of 1.7 and 1.8°C, respectively. A 256 similar approach was used by Arnell et al. (2021), who accept the observed rise of 0.61° C 257

between pre-industrial and 1981–2010 as fixed, then use a projected further increase of
1.39°C relative to 1981–2010 to define the 2°C GWL for each model.

Figure 2 uses the driving (global) models to demonstrate the problems inherent in 260 comparing GWLs with a fixed reference period. For regional model output however, the 261 equivalent of Figure 2c often cannot be produced because, as noted above, regional sim-262 ulations are typically unavailable for time periods before 1980. If the intention of an anal-263 ysis really is to characterise the change in regional climate between 1981–2010 and some 264 future GWL, then one simple approach would be to replace the fixed 1981–2010 refer-265 ence period with a model-dependent reference period defined, for each model, as the time 266 at which the driving GCM reaches a GWL equivalent to that observed in the real world 267 by 1980, and to use this as the basis for comparisons. However, the choice of reference 268 GWL is constrained by the available data: one of the CMIP5-EC models has warmed 269 by 0.8° C prior to the start of the reference period, corresponding to the warming actu-270 ally observed by around 2006. In the following analysis therefore, we explore the range 271 of changes simulated in each of the regional ensembles between periods in which the driv-272 ing GCMs reached GWLs of $1^{\circ}C$ – approximately the level observed by 2015 – and $2^{\circ}C$ 273 respectively. GWL climatologies are calculated by identifying the year in which the GMST 274 of the driving GCM first exceeds the GWL of interest; calculating the climatology of the 275 regional model output for the 30-year periods centred on those years; and computing the 276 change between the two. As above, changes in temperature are presented as absolute changes, 277 while changes in precipitation are presented as relative differences with respect to the 278 amount projected after 1°C of global warming. 279

²⁸⁰ 3 Simulation of UK temperatures

281

3.1 Historical biases, 1989–2008

Figure 3a shows maps of the HadUK-Grid mean daily temperature in winter and 282 the mean bias in each ensemble during the evaluation period (1989–2008), with corre-283 sponding plots for summer temperatures in Figure 3b. The CMIP5-EC runs are on av-284 erage around 1°C too cold over much of central England but somewhat too warm at high elevations and, in winter, around much of the coast. This pattern can be attributed to 286 an underlying cold bias in many of the GCMs consistent with that observed over much 287 of western Europe by Vautard et al. (2021), offset by local warm biases due to unresolved 288 topography and blurring between land and sea surface temperatures due to the coarse 289 resolution. A similar spatial pattern is seen in the GC3.05-PPE ensemble mean, although 290 with much reduced biases at higher elevations. The RCMs inherit this cold bias but are 291 able to resolve the features causing local warm biases in the driving models, with the Eu-292 roCORDEX ensemble as a whole having a fairly uniform bias of between -1 and $-2^{\circ}C$ 293 across the UK land surface in winter (-1°C in summer). In the evaluation runs driven 294 by ERA-Interim reanalysis, the magnitude of this bias is reduced in both summer and 295 winter, supporting the suggestion that the error is to some extent inherited from the driv-296 ing GCMs. The UKCP regional ensemble also inherits a slight cold bias from the 60km 297 driving runs, largely attributed by Murphy et al. (2018) and Tucker et al. (2021) to a 298 strong aerosol forcing, moderated by differences in large-scale circulation patterns; in win-200 ter this bias is slightly smaller on average than seen in the EuroCORDEX ensemble, with 300 a fairly uniform mean bias of -0.5° C across much of the UK, increasing to -1.5° C over 301 higher elevations in Scotland. 302

The boxplots in Figure 4 show the distributions of average UK winter and summer temperatures in each ensemble. Average UK temperatures in the regridded CMIP5-EC ensemble are slightly higher than in the corresponding EuroCORDEX runs, with the differences particularly pronounced in winter (panel a); this is largely due to the warm biases at high elevations and in coastal regions mentioned above, which the regional models are able to resolve. Within the EuroCORDEX ensemble, average summer temper-



Figure 3. Maps of seasonal averages of HadUK-Grid daily mean temperature (in °C) from 1989 to 2008, and of the mean climatological biases in (a) winter and (b) summer, in each of the ensembles of models. The mean bias over the UK land surface is given in parentheses.

atures also display a degree of clustering by RCM, with the coolest summers simulated 309 by runs downscaled using RACMO22E and RCA4 (coloured yellow and lime green); the 310 same RCM ordering is also seen in the reanalysis-driven ERA-EuroCORDEX ensemble, 311 suggesting that the regional models also contribute systematic differences of their own 312 (Sørland et al., 2018; Vautard et al., 2021). UKCP18 ensemble members display simi-313 lar biases at both 60km and 12km resolution and, with the exception of the coldest runs 314 in winter, the spread of biases in the UKCP18 ensembles is broadly comparable to that 315 of the EuroCORDEX ensemble. In both the EuroCORDEX and UKCP regional ensem-316 bles the biases in mean temperatures are largely due to underestimation of daily max-317 ima arising from large-scale processes driven by the GCMs (Vautard et al., 2021), while 318 daily minima are typically well represented. 319

320

3.2 Projected changes in temperature, 2050–2079 relative to 1981–2010

Maps of ensemble mean changes (shown in Figure S1 in the supplementary mate-321 rial, and also available from the accompanying Plot Explorer as detailed in Section 1), 322 indicate a fairly uniform increase in temperature across the whole of the UK, although 323 the UKCP regional ensemble warms somewhat more over higher elevations in winter and 324 in southern England in summer. The distributions of the average changes in seasonal 325 temperature across the UK projected by individual runs are shown in the boxplots in 326 Figure 5. In both winter and summer, EuroCORDEX runs denoted by the same sym-327 bol (indicating that they were driven by the same GCM) are closely grouped together, 328 with the average changes in the EuroCORDEX runs generally of similar magnitude to 329 the changes in the driving GCM runs, indicating that the dominant contribution arises 330 from the driving models: the CMIP5-EC and EuroCORDEX ensembles warm by, on av-331 erage, around 2°C in both winter and summer. This is also the case in the UKCP18 en-332



Figure 4. Boxplots showing the distribution of UK-averaged daily mean temperatures in each ensemble during the evaluation period (1989-2008) during (a) winter and (b) summer months. The boxes indicate the central 50% of the distribution; the whiskers of the boxplot extend to values lying 1.5 times the interquartile range beyond the upper and lower quartiles. The shaded region behind each boxplot shows a kernel density estimate of the empirical distribution of the values. Members of the CMIP5-EC and EuroCORDEX ensembles are represented by coloured symbols, with the shape indicating the GCM used to force the run, and the colour indicating the downscaling RCM; points corresponding to the output of a single GCM are jittered horizontally for ease of viewing. The unperturbed UKCP18 ensemble member, corresponding to HadREM3-GA7-05 in the regional ensemble, is shaded orange.

- sembles, where each ensemble member warms by the same amount at both 60km and 333 12km resolutions: in winter, by around 0.6° C more on average than the EuroCORDEX 334 ensemble, and in summer, by around 1.7°C more. Similar differences between the GC3.05-335 PPE and CMIP5 projections have been discussed by Yamazaki et al. (2021), who attributed 336 them partly to greater climate sensitivity in the UKCP18 members than in most of the 337 CMIP5-EC models, and partly to the fact that the CO_2 pathways sampled by GC3.05-338 PPE tend to lie above the standard RCP8.5 pathway used to drive the CMIP5 runs, as 339 discussed in Section 2.1. Boé et al. (2020) and Taranu et al. (2023) also note that the 340 absence of time-varying aerosols from most of the EuroCORDEX RCM simulations may 341 also suppress the range of future projections in that ensemble. 342
- Readers may note that GC3.05-PPE is derived from a model descended from HadGEM2-343 ES (represented by a cross in the CMIP5-EC and EuroCORDEX ensembles), which also 344 projects a strong warming trend. While the similarity between their projected warm-345 ing levels suggests that these two GCMs share a similar degree of climate sensitivity, the 346 differences between the two models are substantial (Williams et al., 2018; Murphy et al., 347 2018), and the intermediate variant HadGEM3-GC2 introduced changes that reduced 348 climate sensitivity (Senior et al., 2016), before changes to parameterisation (notably in 349 the aerosol and cloud microphysics schemes) increased the sensitivity of the GC3 gen-350 eration of models (Bodas-Salcedo et al., 2019). This greater sensitivity appears to be com-351 pounded during the summer months by the use of HadREM3-GA7-05 (indicated by or-352 ange symbols in the EuroCORDEX ensemble), which produces the warmest run in the 353



Figure 5. Boxplots showing the average change in daily mean (a) winter and (b) summer temperatures over the UK land surface between the reference period (1981–2010) and the future period (2050–2079). For details of the plot elements see the caption to Figure 4.

EuroCORDEX ensemble for every GCM with which it is paired. This difference is driven 354 by particularly large increases in summer daily maxima, which are typically 1°C higher 355 across the UK than the corresponding increases in summer minima in all UKCP regional 356 and HadREM3-GA7-05 runs (Figure S2), probably largely due to the inclusion of aerosol 357 forcing in the regional model (Boé et al., 2020; Tucker et al., 2021). As noted by Lo et 358 al. (2020) and Keat et al. (2021), UKCP regional also exhibits a particularly strong ur-359 ban heat island effect, with summer daily minima in London increasing by around 0.2° C 360 more than in the rest of south-eastern England. Maps and boxplots of biases and changes 361 in seasonal temperature maxima and minima are also available through the online Plot 362 Explorer tool. 363

364 365

3.3 Projected changes in temperature, 2°C relative to 1°C global warming

Figure 6 shows boxplots of the seasonal changes in temperature simulated by the 366 two regional ensembles in response to an increase of GMST from 1° C to 2° C in the driv-367 ing models. During the winter months both the CMIP5-EC/EuroCORDEX and UKCP18 368 ensembles warm by slightly less than 1°C over the UK (0.8°C and 0.6°C respectively, 369 Figure 6a). The UKCP18 ensembles both have a strongly bimodal distribution, with four 370 of the runs warming very little, and the remaining eight runs warming by 0.7-1°C, roughly 371 in line with the central 50% of the EuroCORDEX distribution. This is a contrast to Fig-372 ure 5a, where more than half of the UKCP18 runs exceeded the 75th percentile of the 373 EuroCORDEX ensemble. Even after removing variation associated with different global 374 warming rates, EuroCORDEX runs with the same symbol (denoting the same driving 375 model) are still loosely grouped together, reflecting the importance of large-scale pro-376 cesses in determining daily temperatures over the UK (Pope et al., 2022). 377

The pattern of changes in summer temperatures in Figure 6b bears more resemblance to that seen in Figure 5b: in UKCP18 the UK warms slightly more rapidly than the global mean in summer, with most members simulating increases of 1–1.2°C, while the range of responses simulated by the EuroCORDEX ensemble is rather wider and slightly lower, and broadly similar to the range of responses in winter temperatures. As noted previously, this larger change in summer temperatures in the UKCP18 runs is driven primarily by an increase in daily maxima, with ten of the twelve runs simulating increases of 1.2–1.7°C, while daily minima warm by 0.8–1°C. This difference between the two ensembles can no longer be attributed to a difference in climate sensitivity or in the CO2 pathways sampled, but is still largely determined by the GC3.05-PPE global models.



Figure 6. Boxplots of changes in (a) winter and (b) summer UK mean temperature in response to an increase of GMST from 1°C to 2°C. For details of the plot elements see the caption to Figure 4.

³⁸⁸ 4 Simulation of UK precipitation

389

4.1 Historical biases, 1989–2008

Figure 7 shows maps of the mean winter and summer daily precipitation rates in 390 HadUK-Grid, together with the relative mean biases in each ensemble. Observed pre-391 cipitation is highest in west-facing areas of high elevation throughout the year, with the 392 heaviest rainfall concentrated in western Scotland. In winter, with the exception of the 393 UKCP regional ensemble, the models tend to overestimate precipitation in the drier lower-394 lying areas, and to underestimate it in the wetter areas and at higher elevations. Although 395 the biases are most acute in CMIP5-EC and GC3.05-PPE – indicating that this is due 396 to unresolved features – the fact that they persist in the EuroCORDEX ensembles, both 397 those driven by GCM outputs and by reanalyses, suggests that the RCMs also do not 398 fully resolve the driving processes despite better representing the local topography. For 399 the EuroCORDEX and UKCP regional ensembles, separate analyses of precipitation fre-400 quency and wet-day intensity (Figures S3 and S4, also available through the online Plot 401 Explorer tool: a wet day is one with at least 1mm precipitation) reveals that the wet bi-402 ases in lower-lying, drier regions tend to correspond to simulation of too many wet days, 403 with dry biases at higher elevations the result of lower rates of wet-day precipitation, al-404 though the signs are more mixed in winter: similar results were presented in Kendon et 405 al. (2021). The UKCP regional ensemble does not suffer from this dry elevation bias, and 406



Figure 7. Maps of HadUK-Grid daily mean precipitation rates from 1989 to 2008, and of the relative biases in the means of each ensemble, during (a) winter and (b) summer.

is too wet across much of the UK in winter, with the exception of the far northwest where
 observed precipitation is highest.

In summer the CMIP5-EC and GC3.05-PPE ensembles are also too dry at higher 409 elevations – again, predominantly the result of unresolved topography – while the main 410 EuroCORDEX ensemble is, on average, too wet across most of the UK; the average bias 411 in the reanalysis-driven runs is similar to that seen in winter, but the magnitude is rather 412 smaller. Aside from the elevation-induced biases, the GC3.05-PPE runs are slightly too 413 wet in rain shadows, but too dry across much of England: the UKCP regional runs are 414 slightly too dry in England but again slightly too wet across much of Scotland. Again, 415 inspection of plots available on the Explorer tool indicates that wet-day precipitation rates 416 are typically underestimated across the UK, with too many wet days simulated on av-417 erage in all ensembles except for the GC3.05-PPE ensemble. 418

The distributions of the relative biases in UK mean summer and winter precipi-419 tation within each ensemble are shown in Figure 8. The CMIP5-EC runs underestimate 420 precipitation over the UK in both summer and winter, with most of the models under-421 estimating winter precipitation by more than 20%. This pattern of biases is not directly 422 reflected in the EuroCORDEX biases, which vary widely, ranging from -40% to +50%423 in winter and -30% to +70% in summer. Biases in the EuroCORDEX runs tend to be 424 more closely grouped by colour (denoting the RCM) than by shape (denoting the GCM), 425 and to be similar to those of the corresponding reanalysis-driven ERA-EuroCORDEX 426 runs; this suggests that biases in precipitation are determined to a greater extent by the 427 choice of RCM than the choice of GCM, which in turn implies that the differences be-428 tween the properties of RCM and GCM output are due to more than just the differing spatial resolutions of the models. Biases in average UK precipitation are more closely 430 correlated with biases in the wet-day precipitation rate in winter, and with biases in the 431 number of wet days in summer. 432



Figure 8. Boxplots showing the distribution of relative biases in UK-averaged winter and summer precipitation in each ensemble during the evaluation period (1989-2008). For details of the plot elements see the caption to Figure 4.

Both of the UKCP18 ensembles have a much smaller range of relative biases than 433 the EuroCORDEX ensemble, with the 12km runs being somewhat wetter overall than 434 their 60km driving runs, largely due to compensation of local wet and dry biases in the 435 lower-resolution runs. This difference is primarily due to the ability of the regional mod-436 els to represent orographic processes at finer resolution, and so to reduce the dry bias 437 at higher elevations, although it is that the inclusion of time-varying aerosols in the UKCP 438 regional runs also plays a part (Boé et al., 2020; Tucker et al., 2021). In both summer 439 and winter the relative bias in the mean of the UKCP regional ensemble is similar to the 440 biases in EuroCORDEX runs using HadREM3-GA7-05, the model from which the UKCP 441 regional PPE was constructed, further reinforcing the role of the choice of RCM in de-442 termining biases in precipitation. 443

444

4.2 Projected changes in precipitation, 2050–2079 relative to 1981–2010

Maps of the relative changes in the ensemble means of precipitation between 1981– 445 2010 and 2050–2079 (Figure S5, also available from the accompanying Plot Explorer tool) 446 show an overall increase in winter and a decrease in summer across nearly the entire ter-447 ritory: in winter, the CMIP5-EC and EuroCORDEX ensembles the mean increases are 448 around 10% across much of the UK, while the two UKCP18 ensembles exhibit slightly 449 larger increases on average and a slight gradient, with the 60km ensemble simulating 15-450 25% more precipitation in south-west England and little or no change in the north-east 451 and Scotland. This pattern is also apparent in the 12km ensemble mean, with the in-452 creases slightly damped at higher elevations. In summer, means of the the CMIP5-EC 453 and EuroCORDEX ensembles project around 10% less precipitation across the UK on 454 average, and in the UKCP18 ensembles, an average of 25% less. The UKCP18 ensem-455 bles again display a pronounced northeast-southwest gradient, with as much as 45% less 456 precipitation in the southwest of England; in the 12km ensemble, there is additional dry-457 ing on western-facing elevations and slightly less in rain shadows. 458

The boxplots in Figure 9 show the distribution of the percentage changes in UKaveraged winter and summer precipitation within each ensemble. The trend of increas-

ing mean winter precipitation is fairly consistent across the ensembles: all of the CMIP5-461 EC runs and most of the EuroCORDEX runs project increases of 5-15% in winter pre-462 cipitation (Figure 9a), although a handful of runs from HIRHAM5 and HadREM3-GA7-463 05 simulate less precipitation than during the reference period; all but one of the UKCP18 runs simulate increases of 5-20% at both resolutions. This change is driven primarily by 465 an increase in the wet-day precipitation rate, with very little change in the average num-466 ber of wet days simulated on average across the UK (see also Kendon et al. (2021), who 467 investigate the issue in more detail for the UKCP local ensemble). Within the EuroCORDEX 468 ensemble, points are loosely grouped by shape, indicating that the GCMs are dominant 469 in determining the change in winter precipitation: however, within these groups the points 470 are also ordered by colour, suggesting that the choice of RCM also plays a fairly signif-471 icant part. Again, it is interesting to note that the unperturbed member in the UKCP 472 regional ensemble – produced by the same parametrisation of HadREM3-GA7-05 as the 473 EuroCORDEX runs, and highlighted in orange in the plots – simulates one of the small-474 est increases in winter precipitation in that ensemble, suggesting that this may be a char-475 acteristic of that particular RCM. 476



Figure 9. Boxplots showing the relative changes in accumulated UK precipitation in each ensemble during (a) the winter and (b) the summer months between the reference period (1981–2010) and the future period (2050–2079). For details of the plot elements see the caption to Figure 4.

The distributions of changes in mean summer precipitation in the CMIP5-EC and 477 EuroCORDEX ensembles (Figure 9b) are fairly skewed, with most of the GCMs sim-478 ulating 5-20% less precipitation but with one outlying model – CNRM-CM5 – simulat-479 ing 7.5% more precipitation across the UK than in the reference period: six of the eight 480 EuroCORDEX runs that simulate an increase in summer precipitation across the UK 481 are driven by this GCM, which is the only one in the ensemble to simulate an increase 482 in the number of wet summer days, suggesting that this may be the result of changes 483 in large-scale circulation patterns. The largest reduction in summer precipitation in the 484 CMIP5-EC ensemble is produced by HadGEM2-ES, the model that also simulated the 485 largest increase in summer temperatures (Figure 5b). However, this tendency is not in-486 herited directly by the runs driven by that model, which produce a wide spread of changes 487 in precipitation, including both the largest decrease and the second largest increase in 488

the EuroCORDEX ensemble: as already noted, the choice of RCM also contributes sig-489 nificantly to the differences between individual runs. All of the UKCP18 runs project 490 reductions of at least 10% in summer precipitation across the UK – a stronger drying 491 trend than the EuroCORDEX ensemble mean or median – with an average reduction 492 of 22.5%. Changes in average precipitation during the summer months are driven by a 493 reduction in the number of wet days simulated, with the effect slightly mitigated by small 494 increases in the wet-day precipitation rate in the CMIP5-EC and EuroCORDEX ensem-495 bles, but compounded in several of the UKCP18 runs by small decreases in the rate of 496 wet-day precipitation. This is again consistent with the findings of Kendon et al. (2021)497 and of Pope et al. (2022), who note a projected increase in occurrences of large-scale cir-498 culation patterns associated with dry, settled weather over the UK during the summer 100 months in GC3.05-PPE. 500



4.3 Projected change in precipitation, 2° relative to 1° global warming



Figure 10. Boxplots of changes in (a) winter and (b) summer UK mean precipitation in response to an increase of GMST from 1°C to 2°C. For details of the plot elements see the caption to Figure 4.

502

The relative changes in mean UK winter and summer precipitation in response to an increase in GMST from 1°C to 2°C are shown in Figure 10. The UKCP18 runs sim-503 ulate very little change in winter precipitation on average (panel a), with individual runs 504 projecting between $\pm 5\%$; more than 25% of the EuroCORDEX members also project 505 a reduction of up to 5%, with the remainder projecting increases of up to 16%, slightly 506 lower than the increases seen in Figure 9a. Overall, both ensembles simulate an increase 507 of around 2.5% in wet-day precipitation rates in response to 1°C of continued warming; 508 the EuroCORDEX runs simulate 1-2% more wet days on average, while the UKCP18 509 runs simulate 2.5% fewer, leading to very little net change in precipitation. 510

During the summer months the trends are again similar to those seen in Figure 9b, 511 with EuroCORDEX runs projecting changes from -12% to +8%, and the UKCP18 en-512 semble slightly more intense drying (Figure 10b). However, this difference between the 513 two ensembles is largely driven by just two of the UKCP18 runs, with the remainder span-514 ning the central 70% of the EuroCORDEX ensemble. As noted previously, these changes 515 are largely driven by a reduction in the number of wet days simulated in both ensem-516

⁵¹⁷ bles (by about 5%), with the UKCP18 runs also simulating about 2% less precipitation ⁵¹⁸ on wet days: plots illustrating these changes can be found in Figures S6 and S7 or us-⁵¹⁹ ing the aforementioned Plot Explorer tool. Again we find that, although variability at-⁵²⁰ tributable to differing warming rates has been removed, runs driven by the same GCM ⁵²¹ (denoted by the same symbol) still tend to be grouped together, indicating that the choice ⁵²² of GCM still determines the simulated climate to a large extent.

523 5 Discussion and conclusions

Sections 3 and 4 present an analysis of biases and changes in summer and winter 524 temperatures and precipitation over the UK: while the results presented are specific to 525 the local climate, they offer a useful illustration of the insights that can be gained by con-526 sidering changes over GWLs alongside those over fixed time periods. We anticipate that 527 this analysis could be used as a template for regional ensemble comparisons more widely, 528 providing a framework by which the effect of regional responses to global warming might 529 be assessed alongside projected changes at a given time period, in order to disentangle 530 the drivers. 531

Both the EuroCORDEX and UKCP18 regional model ensembles were found to exhibit a similar range of biases in temperature in both summer and winter; a persistent cold bias is inherited by all runs from the driving models but the regional models reduce this tendency somewhat, and the resulting temperature biases are small on average. The UKCP18 runs tend to be too wet on average in winter, while most of the EuroCORDEX runs are too wet in summer, with wet biases generally associated with the simulation of too many wet days.

When considering changes in local climate over time, the CMIP5-EC/EuroCORDEX 539 and GC3.05-PPE/UKCP regional ensembles generally agree on the sign of the changes 540 in average temperatures and precipitation over the UK; however, a stronger signal is ob-541 served in the UKCP18 runs at both 60km and 12km resolutions, which project much larger 542 temperature increases and larger drying (wettening) effects in the summer (winter) months. 543 This is, in part, due to the fact that the CMIP5-EC runs used to drive the EuroCORDEX 544 simulations do not include the warmest and driest of the CMIP5 projections (Boé et al., 545 2020; Coppola et al., 2021), while GC3.05-PPE is derived from a model known to ex-546 hibit a high rate of warming in response to greenhouse gas emissions (Murphy et al., 2018; 547 Andrews et al., 2019), as illustrated in Table 1. This greater sensitivity is compounded 548 by the use of perturbed CO₂ pathways to force the runs, which resulted in a higher ef-549 fective forcing than the standard RCP8.5 scenario in the majority of ensemble members 550 (Sexton et al., 2021; Yamazaki et al., 2021). 551

The effect of these differences is substantial: between 1981–2010 and 2049–2079, 552 the GC3.05-PPE runs warm by 3-4°C globally; of the ten EuroCORDEX driving runs, 553 HadGEM2-ES and IPSL-CM5A-MR warm by around 3°C, while the remainder warm 554 by around 2.25°C in the same period. Taking into account the number of replicates of 555 each GCM in the EuroCORDEX ensemble, the average global warming across the UKCP 556 regional ensemble will be around 1°C more than the corresponding average for EuroCORDEX 557 during this time: given that changes in many key climate indices have been found to in-558 crease monotonically with GMST change (James & Washington, 2013; Seneviratne & 559 Hauser, 2020), the UKCP regional ensemble should be expected to display a correspond-560 ingly stronger change signal. 561

This divergence between the two ensembles poses a problem for anyone wishing to use these climate projections to support effective planning and decision making: how should the two sets of projections be interpreted? The results presented here indicate that there is no direct relationship between the biases exhibited during the evaluation period and future rates of warming, so simple bias correction methods are unlikely to be able to rec-

oncile the two ensembles. By comparing model outputs at fixed warming levels as in Fig-567 ures 6 and 10, rather than at fixed time periods as in Figures 5 and 9, differences between 568 the EuroCORDEX and UKCP regional ensembles attributable to the varying rates of 569 GMST change in the driving models have largely been removed. As a result the two en-570 sembles, when taken together, present a more coherent picture of plausible local changes 571 in response to global warming, with the UKCP regional ensemble exploring the warmer, 572 drier scenarios that are known to be absent from the EuroCORDEX ensemble (Boé et 573 al., 2020): this complementary information may be important in the context of climate 574 change in western Europe and the UK, where models have been found to underrepre-575 sent observed trends in warming on the warmest summer days (Vautard et al., 2023). 576

Given the current focus on the adaptation to a world 1.5° C or more warmer than 577 the preindustrial climate, this GWL-based analysis has potential applications in sepa-578 rating analysis of the local and regional changes that are to be expected at a given level 579 of global warming from consideration of rates of GMST change. The GWL approach can 580 help to answer the question of why the two ensembles indicate different climate futures, 581 but could also be used to investigate broader questions around adaptation: for exam-582 ple, to what extent are local responses to global warming dependent on the emissions 583 scenario used, the climate sensitivity of the driving models, or the absolute level of warm-584 ing reached? 585

Further work is also required to evaluate the sensitivity of the GWL approach to 586 the time periods compared. Time slices spanning a fixed number of years either side of 587 a given threshold exceedance will contain different ranges of GMSTs depending on the 588 climate sensitivity of the driving models, which may introduce biases, particularly in any 589 indices measuring extrema or variability. Alternative approaches might be to select a sym-590 metric GMST interval centred on the year of interest; or more sophisticated approaches 591 based on detrended residuals during the chosen time period, following an approach sim-592 ilar to that used by Sexton et al. (2012) in a slightly different context. Furthermore, al-593 though the GWL approach reduces some of the discrepancies between the ensembles, it does not fully reconcile them in all respects (see Figures 6 and 10, for example). Some 595 of the reasons for this are outlined above, but these results nonetheless serve as a note 596 of caution that the approach cannot be regarded as a universal panacea, and users should 597 assess the advantages and disadvantages of the approach relative to other frameworks 598 for addressing variability and biases within ensembles of climate projections. 599

In contexts where timescales are important, the information provided by analysis 600 of changes between GWLs may be less directly relevant. Similarly, for indices of quan-601 tities that are less directly dependent on global temperature change – for example, some 602 indices of precipitation, which may be more sensitive to changes in atmospheric circu-603 lation and composition than those determined by temperature – the GWL approach may 604 be less effective in reducing inter-model differences: since different models reach the same 605 GWL at different CO_2 levels, they do so under potentially quite different atmospheric 606 compositions, although some studies have found a monotonic or even linear relationship 607 between regional changes and increasing GMST (James & Washington, 2013; Seneviratne 608 & Hauser, 2020; Arnell et al., 2021). Whether the GWL approach is appropriate or not 609 in a given application, there is still useful information to be gained by comparing the out-610 puts of more than one ensemble of models. 611

One perspective is that ensembles, like the UKCP regional ensemble, with higher 612 warming rates explore the upper tails of the distribution of plausible outcomes, provid-613 ing a set of storylines of low-likelihood but high-impact futures (Zappa & Shepherd, 2017) 614 615 for use in risk-averse decision making. However, neither the EuroCORDEX ensemble nor the combined EuroCORDEX-UKCP regional ensemble systematically samples a range 616 of climate sensitivities, so neither should be interpreted as representative of the possi-617 ble distribution of future scenarios, although the two ensembles taken together are ar-618 guably more representative than either one in isolation. To gain a fuller understanding 619

of the uncertainty about projected changes, it may be instructive to place the regional 620 model output within the context of the UKCP18 probabilistic models, which are designed 621 to more fully reflect the spread of potential future outcomes, or the full UKCP18 global 622 ensemble, which includes not only the PPE but also a subset of thirteen CMIP5 mod-623 els chosen to reflect a wider range of plausible futures (Murphy et al., 2018); both of these 624 products provide global data, although projections are available for fewer climate vari-625 ables and at coarser spatial and temporal resolution than the regional model output, and 626 may therefore not provide sufficient detail for some applications. Recent work has shown 627 that observational constraints accounting for the rate of warming in recent decades can 628 resolve much of the difference between the rates in CMIP5 and CMIP6 (Brunner et al., 629 2020; Ribes et al., 2021), suggesting that similar approaches might be applied to resolve 630 the differences between CMIP5 and GC3.05-PPE, although the method has not been ap-631 plied to maps of the outputs from regional climate models. This problem of how to in-632 terpret and extract relevant information from ensembles that include models with a wide 633 range of climate sensitivities is to become increasingly important, given the known pre-634 ponderance of high-sensitivity models in the CMIP6 ensemble (Zelinka et al., 2020). 635

636 Open Research

All plots and data used in this analysis can be downloaded from the EuroCORDEX-UK plot explorer tool at https://github-pages.ucl.ac.uk/EuroCORDEX-UK-plot-explorer/, along with plots of other climate indices (Barnes et al., 2023).

CMIP5 and EuroCORDEX climate simulations can be obtained from the Earth
 System Grid Federation portals (e.g., https://esg-dnl.nsc.liu.se/search/cordex/,
 https://esgf-node.llnl.gov/search/cmip5/).

All UKCP18 climate simulations can be downloaded from the Centre for Environ mental Data Analysis at https://catalogue.ceda.ac.uk (Met Office Hadley Centre,
 2018), along with EuroCORDEX simulations regridded to the 12km OSGB grid used
 in this report (Barnes, 2023).

647 Acknowledgments

This research was funded under the UK Climate Resilience Programme, which is supported by the UKRI Strategic Priorities Fund. The programme is co-delivered by the Met Office and NERC on behalf of UKRI partners AHRC, EPSRC, and ESRC.

The authors acknowledge the World Climate Research Programme's Working Group 651 on Regional Climate, and the Working Group on Coupled Modelling, former coordinat-652 ing body of CORDEX and responsible panel for CMIP5. We also thank the climate mod-653 elling groups (listed in Figure 1 of this paper) for producing and making available their 654 model output. We also acknowledge the Earth SystemGrid Federation infrastructure, 655 an international effort led by the U.S. Department of Energy's Program for Climate Model 656 Diagnosis and Intercomparison, the European Network for Earth System Modelling and 657 other partners in the Global Organisation for Earth System Science Portals (GO-ESSP). 658

The authors would also like to thank the two anonymous reviewers for their constructive feedback.

661 References

Andrews, T., Andrews, M. B., Bodas-Salcedo, A., Jones, G. S., Kuhlbrodt, T., Manners, J., ... others (2019). Forcings, feedbacks, and climate sensitivity in
 HadGEM3-GC3.1 and UKESM1. Journal of Advances in Modeling Earth
 Systems, 11(12), 4377–4394.

666	Arnell, N. W., Freeman, A., Kay, A. L., Rudd, A. C., & Lowe, J. A. (2021). Indi-
667	cators of climate risk in the UK at different levels of warming. Environmental
668	Research Communications, $3(9)$, 095005 .
669	Barnes, C. (2023). EuroCORDEX-UK: Regional climate projections for the
670	UK domain at 12 km resolution for 1980-2080 [dataset]. NERC EDS
671	Centre for Environmental Data Analysis. Retrieved from https://
672	catalogue.ceda.ac.uk/uuid/b109bd69e1af425aa0f661b01c40dc51 doj:
673	10 5285/b109bd69e1af425aa0f661b01c40dc51
674	Barnes C. Alegre B. Ho-Lyn A. Quinn I. Chandler B. & Brierley C. (2023)
674	August) Fare CORDEY UK relat emplorer Batriaved from https://github
675	August). EuroconDEA-OK piot explorer. Retrieved from https://github
676	
677	Bodas-Salcedo, A., Mulcany, J., Andrews, T., Williams, K., Ringer, M., Field, P., &
678	Elsaesser, G. (2019). Strong dependence of atmospheric feedbacks on mixed-
679	phase microphysics and aerosol-cloud interactions in HadGEM3. Journal of
680	Advances in Modeling Earth Systems, 11(6), 1735–1758.
681	Boé, J., Somot, S., Corre, L., & Nabat, P. (2020). Large discrepancies in summer cli-
682	mate change over Europe as projected by global and regional climate models:
683	causes and consequences. Climate Dynamics, $54(5)$, $2981-3002$.
684	Brunner, L., Pendergrass, A. G., Lehner, F., Merrifield, A. L., Lorenz, R., & Knutti,
685	R. (2020). Reduced global warming from CMIP6 projections when weighting
686	models by performance and independence. Earth System Dynamics, 11(4),
687	995–1012.
688	Chen, D., Rojas, M., Samset, B., Cobb, K., Diongue Niang, A., Edwards, P.,
689	Tréguier, AM. (2021). Framing, Context, and Methods [Book Section]. In
690	Climate Change 2021: The Physical Science Basis, Contribution of Working
691	Group I to the Sixth Assessment Report of the Intergovernmental Panel on
602	Climate Change (p. 147–286) Cambridge United Kingdom and New York
602	NV USA: Cambridge University Press doi: 10.1017/9781009157896.003
695	Christensen O B & Kiellström E (2020) Partitioning uncertainty components
694	of mean climate and climate change in a large ensemble of European regional
695	alimate model projections. Climate Demanica, 1, 16
696	Calling M. Death, D. D. Dhashang, D. Hamis, C. D. Mamba, I. M. Castar
697	D. M. & Webb, M. L. (2011) — Climate model emong foodbacks and fore
698	D. M., & Webb, M. J. (2011). Chimate model errors, feedbacks and forc-
699	ings: a comparison of perturbed physics and multi-model ensembles. Climate
700	Dynamics, 30(9), 1/3/-1/00.
701	Collins, M., Booth, B. B., Harris, G. R., Murphy, J. M., Sexton, D. M., & Webb,
702	M. J. (2006). Towards quantifying uncertainty in transient climate change.
703	Climate dynamics, $27(2)$, 127–147.
704	Coppola, E., Nogherotto, R., Ciarlo', J. M., Giorgi, F., van Meijgaard, E., Kadygrov,
705	$N., \ldots$ others (2021). Assessment of the European climate projections as sim-
706	ulated by the large EURO-CORDEX regional and global climate model ensem-
707	ble. Journal of Geophysical Research: Atmospheres, 126(4), e2019JD032356.
708	Dee, D., Uppala, S., Simmons, A., Berrisford, P., Poli, P., Kobayashi, S., oth-
709	ers (2011). The ERA-Interim reanalysis: configuration and performance of
710	the data assimilation system. Quarterly Journal of the Royal Meteorological
711	$Society, \ 137(656), \ 553-597.$
712	Flato, G., Marotzke, J., Abiodun, B., Braconnot, P., Chou, S. C., Collins, W.,
713	others (2014). Evaluation of climate models. In Climate Change 2013: The
714	Physical Science Basis. Contribution of Working Group I to the Fifth Assess-
715	ment Report of the Interaovernmental Panel on Climate Chanae (np. 741–866).
716	Cambridge University Press.
717	Giorgi F & Gutowski Jr W J (2015) Regional dynamical downscaling and
719	the CORDEX initiative Annual review of environment and resources 10
710	
119	Hausfathar 7 Marval K Schmidt C A Nielson Common I W & Zalinha
720	maustaulier, Δ ., what ver, K ., Schmutt, G. A., Wielsen-Gammon, J. W., & Zellika,

721	M. (2022). Climate simulations: Recognize the 'hot model' problem. Nature
722	Publishing Group.
723	Hollis, D., McCarthy, M., Kendon, M., Legg, T., & Simpson, I. (2019). HadUK-Grid
724	– a new UK dataset of gridded climate observations. Geoscience Data Journal,
725	6(2), 151-159.
726	Iturbide, M., Fernández, J., Gutiérrez, J., Bedia, J., Cimadevilla, E., Díez-Sierra, J.,
727	others (2021). Repository supporting the implementation of FAIR princi-
728	ples in the IPCC-WG1 Atlas. Zenodo. Retrieved from https://github.com/
729	IPCC-WG1/ATLAS/tree/main/datasets-aggregated-regionally/data/
730	$CMIP5/CMIP5_tas_landsea$ doi: $10.5281/zenodo.3691645$
731	Jacob, D., Petersen, J., Eggert, B., Alias, A., Christensen, O. B., Bouwer, L. M.,
732	others (2014). EURO-CORDEX: new high-resolution climate change projec-
733	tions for European impact research. Regional environmental change, $14(2)$,
734	563–578.
735	James, R., & Washington, R. (2013). Changes in African temperature and precipita-
736	tion associated with degrees of global warming. Climatic change, 117(4), 859-
737	872.
738	James, R., Washington, R., Schleussner, CF., Rogeli, J., & Conway, D. (2017).
739	Characterizing half-a-degree difference: a review of methods for identifying
740	regional climate responses to global warming targets. Wiley Interdisciplinary
741	Reviews: Climate Change, 8(2), e457.
742	Jones, P. W. (1999). First-and second-order conservative remapping schemes for
743	grids in spherical coordinates. Monthly Weather Review, 127(9), 2204–2210.
744	Keat, W. J., Kendon, E. J., & Bohnenstengel, S. I. (2021). Climate change over
745	UK cities: the urban influence on extreme temperatures in the UK climate
746	projections. Climate Dynamics, $57(11-12)$, $3583-3597$.
747	Kendon, E., Short, C., Pope, J., Chan, S., Wilkinson, J., Tucker, S., Harris, G.
748	(2021). Update to UKCP Local (2.2km) projections (Tech. Rep.). Met Of-
749	fice. Exeter. Betrieved from https://www.metoffice.gov.uk/research/
750	approach/collaboration/ukcp/science/science-reports
751	Lehner, F., Deser, C., Maher, N., Marotzke, J., Fischer, E. M., Brunner, L.,
752	Hawkins, E. (2020). Partitioning climate projection uncertainty with multiple
753	large ensembles and CMIP5/6. Earth Sustem Dunamics. 11(2), 491–508.
754	Lo Y E Mitchell D M Bohnenstengel S I Collins M Hawkins E Hegerl
755	G. C Stott, P. A. (2020). UK climate projections: summer davtime and
756	nighttime Urban Heat Island changes in England's major cities Journal of
757	Climate, 33(20), 9015–9030.
758	Met Office Hadley Centre (2018) UKCP18 Regional Climate Model Projec-
750	tions for the UK [dataset] NERC EDS Centre for Environmental Data
760	Analysis, 2023. Retrieved from http://catalogue.ceda.ac.uk/uuid/
761	b4d24b3df3754b9d9028447eb3cd89c6
762	Morice C P Kennedy J J Bayner N A Winn J Hogan E Killick B
763	Simpson I (2021) An updated assessment of near-surface temperature
764	change from 1850: the HadCBUT5 data set Journal of Geophysical Research:
765	Atmospheres, $126(3)$, e2019JD032361.
766	Murphy I Harris G Sexton D Kendon E Bett P Clark B others
767	(2018) UKCP18 land projections: science report. Met Office
769	Nijsse F. I. Cox P. M. & Williamson M. S. (2020). Emergent constraints on tran-
760	signt climate response (TCR) and equilibrium climate sensitivity (ECS) from
770	historical warming in CMIP5 and CMIP6 models Earth System Dynamics
771	11(3), 737-750.
772	Perry M Hollis D & Elms M (2009) The generation of daily aridded datasets of
773	temperature and rainfall for the UK (Tech Rep.) National Climate Informa-
113	tion Contro Mot Office Evotor
774	

Pope, J. O., Brown, K., Fung, F., Hanlon, H. M., Neal, R., Palin, E. J., & Reid,

776	A. (2022). Investigation of future climate change over the British Isles using
777	weather patterns. Climate Dynamics, 58(9), 2405–2419.
778	Ribes, A., Qasmi, S., & Gillett, N. P. (2021). Making climate projections conditional
779	on historical observations. Science Advances, $7(4)$, eabc0671.
780	Seneviratne, S. I., & Hauser, M. (2020). Regional climate sensitivity of climate ex-
781	tremes in CMIP6 versus CMIP5 multimodel ensembles. Earth's future, $8(9)$,
782	e2019 EF001474.
783	Senior, C. A., Andrews, T., Burton, C., Chadwick, R., Copsey, D., Graham, T.,
784	others (2016). Idealized climate change simulations with a high-resolution
785	physical model: HadGEM3-GC2. Journal of Advances in Modeling Earth
786	$Systems, \ 8(2), \ 813-830.$
787	Sexton, D. M., McSweeney, C. F., Rostron, J. W., Yamazaki, K., Booth, B. B., Mur-
788	phy, J. M., Karmalkar, A. V. (2021). A perturbed parameter ensemble of
789	HadGEM3-GC3.05 coupled model projections: Part 1: selecting the parameter
790	combinations. Climate Dynamics, $56(11)$, $3395-3436$.
791	Sexton, D. M., Murphy, J. M., Collins, M., & Webb, M. J. (2012). Multivariate
792	probabilistic projections using imperfect climate models part I: outline of
793	methodology. Climate dynamics, $38(11)$, $2513-2542$.
794	Sobolowski, S., Somot, S., Fernandez, J., Evin, G., Maraun, D., Kotlarski, S.,
795	Brands, S. (2023, February). EURO-CORDEX CMIP6 GCM selection &
796	ensemble design: Best practices and recommendations. Zenodo. Retrieved from
797	https://doi.org/10.5281/zenodo.7673400 doi: 10.5281/zenodo.7673400
798	Sørland, S. L., Schär, C., Lüthi, D., & Kjellström, E. (2018). Bias patterns and cli-
799	mate change signals in GCM-RCM model chains. Environmental Research Let-
800	ters, 13(7), 074017.
801	Taranu, I. S., Somot, S., Alias, A., Boé, J., & Delire, C. (2023). Mechanisms behind
802	large-scale inconsistencies between regional and global climate model-based
803	projections over Europe. Climate Dynamics, $60(11-12)$, $3813-3838$.
804	Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and
805	the experiment design. Bulletin of the American Meteorological Society, $93(4)$,
806	485 - 498.
807	Tucker, S. O., Kendon, E. J., Bellouin, N., Buonomo, E., Johnson, B., & Murphy,
808	J. M. (2021). Evaluation of a new 12km regional perturbed parameter ensem-
809	ble over Europe. Climate Dynamics, 1–25.
810	Van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard,
811	K., \ldots others (2011). The representative concentration pathways: an overview.
812	$Climatic \ change, \ 109(1), \ 5 ext{}31.$
813	Vautard, R., Cattiaux, J., Happé, T., Singh, J., Bonnet, R., Cassou, C., oth-
814	ers (2023). Heat extremes in Western Europe increasing faster than sim-
815	ulated due to atmospheric circulation trends. Nat Commun, 14. doi:
816	https://doi.org/10.1038/s41467-023-42143-3
817	Vautard, R., Kadygrov, N., Iles, C., Boberg, F., Buonomo, E., Bülow, K.,
818	Wulfmeyer, V. (2021). Evaluation of the Large EURO-CORDEX Re-
819	gional Climate Model Ensemble. Journal of Geophysical Research: At-
820	mospheres, 126(17), e2019JD032344. Retrieved from https://agupubs
821	.onlinelibrary.wiley.com/doi/abs/10.1029/2019JD032344 doi:
822	https://doi.org/10.1029/2019JD032344
823	von Trentini, F., Leduc, M., & Ludwig, R. (2019). Assessing natural variability
824	in RCM signals: comparison of a multi model EURO-CORDEX ensemble
825	with a 50-member single model large ensemble. Climate Dynamics, $53(3)$,
826	1963–1979.
827	Williams, K., Copsey, D., Blockley, E., Bodas-Salcedo, A., Calvert, D., Comer, R.,
828	others (2018). The Met Office Global Coupled Model 3.0 and 3.1 (GC3.0
829	and GU3.1) configurations. Journal of Advances in Modeling Earth Systems,
830	10(2), 357-380.

- WMO. (2017). *WMO guidelines on the calculation of climate normals*. World Meteorological Organization Geneva, Switzerland.
- Yamazaki, K., Sexton, D. M., Rostron, J. W., McSweeney, C. F., Murphy, J. M., &
 Harris, G. R. (2021). A perturbed parameter ensemble of HadGEM3-GC3.05
 coupled model projections: Part 2: global performance and future changes.
 Climate Dynamics, 56(11), 3437–3471.
- Zappa, G., & Shepherd, T. G. (2017). Storylines of atmospheric circulation change
 for European regional climate impact assessment. Journal of Climate, 30(16),
 6561–6577.
- Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi,
 P., ... Taylor, K. E. (2020). Causes of higher climate sensitivity in CMIP6
 models. *Geophysical Research Letters*, 47(1), e2019GL085782.